Author's response to reviews

Title: Explaining the effects of an intervention designed to promote evidence-based diabetes care: a theory-based process evaluation of a pragmatic cluster randomised controlled trial

Authors:

Jillian J Francis (j.francis@abdn.ac.uk)
Martin P Eccles (Martin.Eccles@newcastle.ac.uk)
Marie Johnston (m.johnston@abdn.ac.uk)
Paula Whitty (p.m.whitty@ncl.ac.uk)
Jeremy M Grimshaw (jgrimshaw@ohri.ca)
Eileen F S Kaner (E.F.S.Kaner@newcastle.ac.uk)
Liz Smith (liz.smith@mbs.ac.uk)
Anne Walker (anne_walker@yahoo.co.uk)

Version: 5 Date: 14 November 2008

Author's response to reviews:

In the main document:

Author affiliations: numbers 4, 5, 6 changed to superscript
Page 6: underscore removed from the space after "prediction"
Page 12: Comment deleted
Table 3: reformatted so that asterisks to indicate significance do not wrap around within columns

ALSO: I do apologise, but I have searched this webpage and cannot find where or how I can change the following two details:

Figure 2 caption: needs to add "(bolded, square brackets)" added after "practice-level" as it is currently not clear which numbers refer to individual level and which to practice level.

Acknowledgement: The final sentence needs to be updated as "At the time of writing Eileen Kaner held a Department of Health funded NHS Primary Care Career Scientist award".

Sorry for all these changes.

jill